Author’s response to reviews

Title: Intimate partner violence against women and its association with pregnancy loss in Ethiopia: Evidence from a national survey

Authors:

Tenaw Yimer Tiruye (tenyim09@gmail.com)
Catherine Chojenta (catherine.chojenta@newcastle.edu.au)
Melissa L. Harris (melissa.harris@newcastle.edu.au)
Elizabeth Holliday (Liz.Holliday@newcastle.edu.au)
Deborah Loxton (Deborah.Loxton@newcastle.edu.au)

Version: 2 Date: 20 Jun 2020

Author’s response to reviews:

Dear editor and reviewers,
We would like to thank you for the second round reviews of our manuscript entitled “Intimate partner violence against women and its association with pregnancy loss in Ethiopia: Evidence from a national survey” and for providing us with advice to improve the quality of the manuscript. The responses to the comments have been prepared in a summary table and attached as ‘supplementary material’.
Sincerely,
Tenaw Yimer Tiruye (on behalf of the authors)